Author’s response to reviews

Title: Hepatolithiasis with biliary ascariasis - A case report

Authors:

Dr Kirtikumar S Pilankar (kirtikumarp@yahoo.com)
Dr Anjali D Amarapurkar (anjali_1963@hotmail.com)
Rajeev M Joshi (drrajeevioshi@yahoo.com)
Tilakdas S Shetty (tsshetty@rediff.com)
Amit S Khitani (amit_khithani@yahoo.com)
Vipul V Chemburkar (vipulchem@indiatimes.com)

Version: 6 Date: 24 Aug 2003

Respected Madam/Sir,

I undersigned Dr, Kirtikumar S Pilankar have submitted revised manuscript of 'Hepatolithiasis with biliary ascariasis- A case report'
1. I have made suggested corrections in article.
2. CT Scan plates are attached with the article.
3. I am sorry that we could not send gross photograph as multiple sections were studied to demonstrate parasite.
4. We have given explanation about queries asked in the revised article.
5. Written informed consent taken from patient for publication of details.

We are sorry for delay in sending article.
Kindly do the needful. Thanking you.

Dr. Kirtikumar S Pilankar
kirtikumarp@yahoo.com